Author's response to reviews

Title: Comparison of Two Interferon Gamma Release Assays in the diagnosis of Mycobacterium tuberculosis infection and disease in The Gambia

Authors:

Ifedayo M.O Adetifa (iadetifa@mrc.gm)
Moses D Lugos (mlugos2003@yahoo.com)
Abdulrahman Hammond (ahammond@mrc.gm)
David J Jeffries (djeffries@mrc.gm)
Simon Donkor (sdonkor@mrc.gm)
Richard A Adegbola (radegbola@mrc.gm)

Version: 4 Date: 22 October 2007

Author's response to reviews: see over
Dr. J.A. Le Good
Assistant Editor
BMC-series journals

Dear Dr J. A. Le Good,

Changes to Manuscript: Comparison of two interferon gamma release assays in the diagnosis of Mycobacterium tuberculosis infection and disease in The Gambia

We are delighted that BMC Infectious Diseases will be publishing our article in the very new future.

Please find below a description of the formatting changes in new version of our paper.

- We have used the sentence case in the title both in manuscript and submission system.
- The author list has been corrected in the submission system.
- The website and website addresses in the manuscript have been removed and cited in the main reference list.
- Competing interest- this is now stated appropriately.
- Authors’ contributions-this is now formatted as preferred by the journal.
- Table 2 and figures 1-3 cannot fit onto a portrait A4 page. No changes have been made to their orientation.
- ‘Note’ has been removed from the legend for table 1
- Page 19 has been deleted
Thank you.

Yours sincerely

[Signature]

Ifedayo Adetifa
Philip Hill
On behalf of the authors